Author's response to reviews

Title: A Comprehensive Cohort Model in Orthopaedic Trauma Trials

Authors:

Rebecca S Kearney (r.s.kearney@warwick.ac.uk)
Juul Achten (j.achten@warwick.ac.uk)
Nicholas Parsons (n.parsons.2@warwick.ac.uk)
Matthew L Costa (matthew.costa@warwick.ac.uk)

Version: 2 Date: 24 October 2010

Author's response to reviews: see over
Dear Editor,

Please find attached the revised manuscript entitled ‘A Comprehensive Cohort Model in Orthopaedic Trauma Trials’ for consideration of publication.

‘Track Changes’ has been used to highlight where changes to the manuscript have been made. Most of these are to improve the ‘readability’ of the paper. In addition, below is a point by point response the reviewer’s comment’s:

**Reviewer 1 Comments:**

a. Recommend power analysis and continued recruitment for a larger study.

b. Move first 12 rows from results section to the methods section.

**Response:**

a. The need for a power analysis for a larger study has been acknowledged within the manuscript.

b. The first 12 rows have been moved from the results section to the methods section.

**Reviewer 2 Comments:**

a. Note that the 18 per 100,000 incidence is based upon Scandinavian literature.

**Response:**

a. This has been noted within the manuscript.

Regards

Matthew Costa